Author’s response to reviews

Title: An Investigation of the Relationship Between Autonomy, Childbirth Practices, and Obstetric Fistula Among Women in Rural Lilongwe District, Malawi

Authors:

Julika Kaplan (julika.kaplan@bcm.edu)

Jonathan Kandodo (j.kandodo@gmail.com)

Joseph Sclafani (joseph.sclafani@bcm.edu)

Susan Raine (susan.raine@bcm.edu)

Jennifer Blumenthal-Barby (jennifer.blumenthal-barby@bcm.edu)

Alison Norris (norris.570@osu.edu)

Abigail Norris-Turner (ant@osumc.edu)

Elly Chemey (elly@childlegacy.org)

John Beckham (john.beckham@bcm.edu)

Zara Khan (zk6@rice.edu)

Reginald Chunda (rchunda@gmail.com)

Version: 2 Date: 21 Apr 2017

Author’s response to reviews:

We have removed our cover letter, included a "Consent for Publication" statement in the declarations section, and uploaded a "clean" copy of our manuscript.